Author’s response to reviews

Title: A randomized, double-blind, placebo-controlled trial of calcium acetate on serum phosphorus concentrations in patients with advanced non-dialysis-dependent chronic kidney disease

Authors:

WAJEH QUNIBI (QUNIBI@UTHSCSA.EDU)
WOLFGANG C WINKELMAYER (winkelmayer@stanford.edu)
RICHARD SOLOMAN (Richard.Solomon@vtmednet.org)
MOUSTAFA MOUSTAFA (mamoustafa@hotmail.com)
PAUL KESSLER (pkessler@nabi.com)
CHIANG-HONG HO (ChiangHong.Ho@fmc-na.com)
JONATHAN GREENBERG (jonathan.greenberg@fmc-na.com)
JOSE A DIAZ-BUXO (Jose.Diaz-Buxo@fmc-na.com)

Version: 2 Date: 18 September 2010

Author’s response to reviews:

Trial registration number has been entered as the last line of the abstract of the manuscript